Author's response to reviews

Title: A registry-based follow-up study, comparing the incidence of cardiovascular disease in native Danes and immigrants born in Turkey, Pakistan and the former Yugoslavia: Do social inequalities play a role?

Authors:

Nana F Hempler (nahe@sund.ku.dk)
Finn B Larsen (finn.breinholt@stab.rm.dk)
Signe S Nielsen (s.smith.nielsen@pubhealth.ku.dk)
Finn Diderichsen (f.diderichsen@pubhealth.ku.dk)
Anne H Andreasen (ahea@glo.regionh.dk)
Torben Jorgensen (torjoe01@glo.regionh.dk)

Version: 6 Date: 22 August 2011

Author's response to reviews:

Dear editor,

We have made the few requested corrections. Please notice that we have changed the author list.

Kind regards,

Nana Folmann Hempler